Author's response to reviews

Title: Implementing a Disability Determination and Welfare System Based on the Framework of the International Classification of Functioning, Disability, and Health: The Experience in Taiwan

Authors:

Wen-Ta Chiu (wtchiu.tmu@gmail.com)
Chia-Feng Yen (mapleyeng@gmail.com)
Sue-Wen Teng (nhswteng@doh.gov.tw)
Hua-Fang Liao (hfliao@ntu.edu.tw)
Kwan-Hwa Chang (chang2773@gmail.com)
Wen-Chou Chi (y6312002@gmail.com)
Yen-Ho Wang (lukewang@ntu.edu.tw)
Tsan-Hon Liou (peter_liou@s.tmu.edu.tw)

Version: 2 Date: 27 April 2013

Author's response to reviews: see over
Dear Flory Mae Calumpita

We are submitting a revised article MS: 1162483105975182 Implementing a Disability Determination and Welfare System Based on the Framework of the International Classification of Functioning, Disability, and Health: The Experience in Taiwan. My responses regarding to your comments are listed point-by-point as follows,

1. Ethics statement
   Response: We provided the IRB numbers of this study and added one sentence “This study was approved by the Joint Institutional Review Board at the Taipei Medical University with approval number of 201004001 and 201205042.” in the method section. Please see P7, L19-21 in my revised manuscript.

2. Consent statement
   Response: We didn’t obtained written informed consent from participants because we collected their scramble data from the national registration system. This was approved by the Joint IRB. Please see P7, L17-19.

3. Copyediting
   Response: We asked the Wallace Academic Editing Company to correct our manuscript. Please find the attached English Editing Certificate.

Sincerely yours
Tsan-Hon Liou, MD PhD

Associate Professor, Graduate Institute of Injury Prevention, Taipei Medical University, 250 Wu-Hsing St., Taipei 110, Taiwan

Tel: +886-2-2249-0088 ext. 1600, Fax: +886-2-2248-0577

Email: peter_liou@s.tmu.edu.tw